Author's response to reviews

Title: Traditional and non-traditional treatments for autism spectrum disorder with seizures: an on-line survey.

Authors:

Richard E Frye (Richard.E.Frye@uth.tmc.edu)
Swapna Sreenivasula (Swapna.Sreenivasula@uth.tmc.edu)
James B Adams (Jim.Adams@asu.edu)

Version: 5 Date: 9 May 2011

Author's response to reviews: see over
The BMC Series Editorial Production Team

Dear BMC Series Editorial Production Team:

We greatly appreciate the decision of our manuscript entitled “Traditional and non-traditional treatments for autism spectrum disorder with seizures: an on-line survey.”

We have made edits to the manuscript per the production staff’s request. We have made all major revision, including fixing the Affiliations, Authors’ contributions, References and Tables. In addition, the manuscript has been edited to fix typos and unclear wording. The supplemental tables have been corrected and one of the figures has been corrected for a missing symbol.